Author’s response to reviews

Title: Single tooth torque correction in the lower frontal area by a completely customized lingual appliance

Authors:

Collin Jacobs (collin.jacobs@unimedizin-mainz.de)
Milena Katzorke (katzorkemilena@gmail.com)
Dirk Wiechmann (wiechmann@lingualsystems.de)
Heiner Wehrbein (heiner.wehrbein@unimedizin-mainz.de)
Rainer Schwestka-Polly (Schwestka-Polly.Rainer@mh-hannover.de)

Version: 1 Date: 27 Jun 2017

Author’s response to reviews:

Dear Professor Stamm,

Thank you for the opportunity to resubmit our paper formerly entitled "Single tooth torque correction in the lower frontal area by a completely customized lingual appliance".

We want to thank the reviewers as well for the positive and constructive feedback and took all aspects mentioned by the reviewers into account and made changes to our paper. All changes are marked yellow in the paper itself and in the cover letter.

Yours sincerely

Collin Jacobs